Micro data on nominal rigidity, inflation persistence and optimal monetary policy



by Engin Kara

September 2009 No 175



Editorial Director Jan Smets, Member of the Board of Directors of the National Bank of Belgium

Statement of purpose:

The purpose of these working papers is to promote the circulation of research results (Research Series) and analytical studies (Documents Series) made within the National Bank of Belgium or presented by external economists in seminars, conferences and conventions organised by the Bank. The aim is therefore to provide a platform for discussion. The opinions expressed are strictly those of the authors and do not necessarily reflect the views of the National Bank of Belgium.

Orders

For orders and information on subscriptions and reductions: National Bank of Belgium, Documentation - Publications service, boulevard de Berlaimont 14, 1000 Brussels.

Tel +32 2 221 20 33 - Fax +32 2 21 30 42

The Working Papers are available on the website of the Bank: http://www.nbb.be.

© National Bank of Belgium, Brussels

All rights reserved.

Reproduction for educational and non-commercial purposes is permitted provided that the source is acknowledged.

ISSN: 1375-680X (print) ISSN: 1784-2476 (online)

Abstract

The popular Calvo model with indexation (Christiano, Eichenbaum and Evans, 2005) and sticky information (Mankiw and Reis, 2002) model have guided much of the monetary policy discussion. The strength of these approaches is that they can explain the persistence of inflation. However, both of these theories are inconsistent with the micro data on prices. In this paper, I evaluate the consequences of implementing policies that are optimal from the perspective of models that overlook the micro-data. To do so, I employ a Generalized Taylor Economy (GTE) (Dixon and Kara, 2007). While there is no material difference between the GTE and its popular alternatives in terms of inflation persistence, a difference arises when it comes to the micro-data: the GTE is consistent with the micro-data. The findings reported in the paper suggest that policy conclusions are significantly affected by whether persistence arises in a manner consistent with the micro-data and that policies that are optimal from the perspective of the models that are inconsistent with the micro-data and that policies that are optimal from the perspective of the models that are inconsistent with the micro-data and that policies that are optimal from the perspective of the models that are inconsistent with the micro-data and that policies that are optimal from the GTE.

Key Words: Inflation persistence, DSGE models, Optimal Monetary Policy.

JEL Classification: E1, E3, E52.

Corresponding author:

Engin Kara, NBB Research Department. e-mail: mail@enginkara.eu

I thank Luc Aucremanne, Huw Dixon, Emi Nakamura and Raf Wouters for their comments on a previous draft of this paper. I thank participants at presentations at Birmingham, the National Bank of Belgium, the Bank of Canada and the Reserve Bank of New Zealand for helpful comments. Faults remain my own.

The views expressed in this paper are my own and do not necessarily reflect the views of the National Bank of Belgium.

TABLE OF CONTENTS

1. In	troduction	. 1
2. Tr	ne Model	. 4
2.1.	Structure of the Economy	. 4
2.2.	Log-linearized Economy	. 5
2.3.	Price-Settings Rules	. 7
2.4.	Monetary Policy Rules	. 8
3. W	elfare Function: Woodford's Approximation	. 9
4. Tr	e Choice of Parameters	10
5. In	flation Dynamics and Micro-evidence on Prices	12
5.1.	Discussion	13
6. Re	esults	14
6.1.	Optimal Policy and Micro-evidence on Prices	15
6.2.	Robustness to the Calvo - GTE	18
7. Sı	Immary and Conclusions	19
Refer	ences	21
Appe	ndix	25
Table	s and figures	30
Natio	nal Bank of Belgium - Working papers series	35

1 Introduction

The Calvo model with indexation (i.e. the IC model) (Christiano et al. (2005), Smets and Wouters (2003)) has been a popular approach to monetary policy analysis¹. According to this model, firms set their prices in nominal terms for a random duration, as in the Calvo model, but throughout the contract length the nominal price is updated according to recent inflation (i.e. indexation). The model was developed to better represent inflation dynamics. As is well known, the Calvo model has been inadequate in this respect.

There is, however, a familiar warning that something is wrong with the Calvo with indexation model. The idea that prices are indexed to an inflation index now appears to be a myth. That is, the notion of indexation implies that all firms in the economy continuously adjust their prices but this contradicts micro-evidence on prices². The micro-evidence provided by the European Central Bank's Inflation Persistence Network for the Euro Area indicates that prices remain unchanged for several months³. Findings reported in Bils and Klenow (2004) and Nakamura and Steinsson (2007a) indicate the same conclusion for the US economy.

Therefore, while the assumption of indexation greatly improves the empirical performance of the Calvo model, there is a definite error that is induced through this assumption. Thus, this approach to monetary policy analysis is problematic and any policy recommendation that arises from this model is questionable at best.

The problem under discussion here is not just a matter of theoretical significance but is also a matter of practical importance. Models developed at the European Central Bank provide an excellent demonstration of its importance. These models include the New Area Wide Model (NAWW) (Christoffel, Coenen and Warne (2008)) and the model developed by Christiano, Motto and Rostagno (2008) (CMR). Smets (2008) notes that these models are "routinely used" at the European Central Bank for monetary policy analysis. CMR assume full indexation⁴. The NAWW model esti-

¹See Schorfheide (2008) for a survey.

²see Woodford (2007) and Chari, Kehoe and McGrattan (2008)

³Dhyne, Alvarez, Bihan, Veronese, Dias, Hoffmann, Jonker, Lunnemann, Rumler and Vilmunen (2005) summarise the findings of the IPN.

⁴Note that in the CMR model, a different approach to indexation is adopted. It is assumed that prices are indexed to an "indexation" index, which is a weighted average of

mates that the degree of indexation in the Euro Area is lower than that in the CMR model and is around 40%. However, the reduced degree of indexation comes at a cost: in the NAWW model, the degree of nominal rigidity is much higher than what the micro-evidence on prices suggests. Specifically, according to this model, the proportion of firms that reset their contracts in a quarter is around 8%, whereas the micro-evidence provided by Dhyne et al. (2005) for the Euro Area indicates that this number is around 25%. It is not just that all prices change in each period, but also that the degree of nominal rigidity of the NAWW model is higher than suggested by the micro-evidence.

A question thus arises: what are the consequences of using a model for monetary policy analysis that overlooks the micro-evidence on prices? To answer this question, we require a model that is as successful as the Calvo with indexation in generating inflation persistence and, at the same time, is consistent with the micro-evidence on prices.

The first model that comes to mind is the sticky information (SI) model developed by Mankiw and Reis (2002), which is commonly viewed as a promising tool to replace the Calvo with indexation. In this model, there is an uncertain contract length, as in the Calvo model, and firm set prices for each period at the beginning of the contract, as in Fischer (1977). Therefore, prices are conditional on the information firms have when they set prices, so as the contract grows older information becomes increasingly out of date. The question then, is whether the sticky information model any better the Calvo with indexation; the answer is no. It is not just the Calvo with indexation that is flawed; the sticky information model itself can be misleading for the same reasons. In fact, as Dixon and Kara (2008) argue, the model is similar to the Calvo with indexation in that, like the Calvo with indexation, it can generate inflation persistence and prices change every period. Therefore, the model generates inflation persistence at the cost of having prices change every period and, therefore, is inconsistent with the micro-data.

However, there is an alternative to these models, namely, the Generalized

past inflation and the central bank's time varying inflation objective. Specifically, in this model, the central bank changes its target every period and firms adjust their prices every period according to the central bank's objective. Their findings indicate that the weight on past inflation in such an index is around 10%. Given that number, they conclude that the degree of indexation in their model is low. This conclusion, however, is incorrect, as it ignores the effect of the time varying inflation objective assumption on prices. Regardless of the degree of indexation to past inflation, the prices remain fully indexed in the CMR model since the authors replace like with like.

Taylor Economy (GTE), which is outlined in Dixon and Kara (2007). The GTE generalizes the Taylor model to allow for sectoral heterogeneity with contract lengths suggested by the micro-data. In the GTE, not all prices change in each period consistent with the micro-data on prices. Dixon and Kara (2008) show that the GTE can potentially explain inflation persistence. Indeed, as I will show in section 3, there is no material difference between the GTE and its popular alternatives in terms of inflation persistence. Another desirable feature of the GTE approach is that it is general enough that it can be used to model any distribution of contract lengths including the one generated by the Calvo model. In fact, the framework on which the GTE is built is sufficiently general that it includes all the main approaches of modelling nominal rigidities as a special case. Hence, it provides a platform on which compare the inflation dynamics and monetary policy implications of alternative approaches.

Given that there is no material difference between the GTE approach and its popular alternatives when it comes to the macro-data but there is a difference when it comes to the micro-data, it is safe to conclude that the GTE provides a more plausible explanation of inflation persistence than its popular alternatives. Thus, it should provide more reliable insights about the choices policymakers face.

In this paper, I use the GTE to investigate the consequences of implementing a policy that is optimal from the perspective of the IC or SI models. I first discuss how policy conclusions are affected by whether inertia arises in a manner consistent with micro-evidence. I then consider the case in which the central bank employs a policy that is optimal from the perspective of a model that is inconsistent with the micro-data if the true economy is assumed to follow the GTE.

The conclusions of this paper are briefly summarised as follows: first, the results reported in the paper illustrate the potential for conclusions based on the IC and SI models to be misleading. This is because policy conclusions that arise from these models are significantly affected by the aspects of the models that are inconsistent with the micro-data. Second, the policy rules that are optimal from the perspective of the IC and SI models can lead to a large welfare loss in the GTE.

Section 2 outlines a macroeconomic framework that allows for the exploration of policy implications of the different models within a common environment. Section 3 derives a utility-based objective function for the central bank. Section 4 describes the calibration of the parameters. Section 5 evaluates the impulse response functions of the different price models. Section 6 presents the results. Section 7 concludes the paper.

2 The Model

The framework presented here is based on Dixon and Kara (2008). There, a framework was developed that encompasses all of the main price-setting frameworks. The approach of the model is to consider an economy consisting of many sectors differentiated by how long a contract lasts. When each sector has a Taylor-style contract we have a Generalized Taylor Economy (GTE). When each sector has a Fischer-style contract, we have a Generalized Fischer Economy (GFE). The Mankiw-Reis sticky-information (SI)model is a special case of the GFE. We also allow for indexation.

The exposition here aims to outline the basic building blocks of the model. I will first describe the structure of the contracts in the economy, then the price-setting process under different models and finally monetary policy.

2.1 Structure of the Economy

In this model, as in a standard DSGE model, there are three types of agents: households, the government and firms. Households and the government are both standard new Keynesian. There is a continuum of identical and infinitely lived households ($h \in [0, 1]$). The households derive utility from consumption and leisure. The government conducts monetary policy and levies a proportional tax τ_t on all goods. τ_t follows an AR(1) process⁵. Corresponding to the continuum of households h there is a unit interval of firms, $f \in [0, 1]$. Each firm f is twinned with household h (f = h)⁶. A typical firm is standard new-Keynesian. It has a monopoly power over a specific product, for which the demand has a constant price elasticity θ . It operates a technology, $Y_{ft} = Z_t L_{ft}$, that transforms labour (L_{ft}) into output (Y_{ft}) subject to productivity shocks (Z_t). These products are then combined to produce the final consumption good Y_t . The production function or aggregator is Dixit-Stiglitz.

⁵The tax shock can be considered analogously to supply shocks.

⁶This assumption means that there is a firm- specific labour market. The implications of the firm-specific labour market assumption on inflation dynamics are well known (see for example Woodford(2003, p. 163-178) Dixon and Kara (2007) and Edge (2002)).

Our assumption on the structure of contracts is novel. We divide the unit interval into segments corresponding to sectors and cohorts within sectors. There are N sectors⁷, i = 1...N, with sector shares α_i summing to unity $\left(\sum_{i=1}^{N} \alpha_i = 1\right)$. Contracts in sector *i* last for *i* periods. Within each sector *i*, there are *i* equally sized cohorts of unions and firms: in each period, one cohort comes to the end of its contract and starts a new one. A standard Taylor model is represented by an economy in which one sector (usually with i = 2 or 4) has a share of unity, the rest has zero. In the GTE, in each sector *i* there is a Taylor contract; in the GFE, a Fischer-style contract⁸. The SImodel is a special case of the GFE.

The simple Calvo model is different from the GTE because price setters do not know how long the contract will last: in each period a fraction ω of firms chosen randomly start a new contract. However, the Calvo process can be described in deterministic terms at the aggregate level because firmlevel randomness washes out. As shown in Dixon and Kara (2006a), the distribution of contract lengths across firms is given by $\alpha_i = \omega^2 i (1 - \omega)^{i-1}$: $i = 1...\infty$, with mean contract length $T = 2\omega^{-1} - 1$ The Calvo model with indexation has the same structure in terms of contract lengths, but there is indexation throughout the contract life in response to past inflation.

2.2 Log-linearized Economy

In this section I will simply present the log-linearized macroeconomic framework⁹. The sectoral output level y_{it} can be expressed as a function of the sectoral price p_{it} relative to the aggregate price level p_t and aggregate output y_t , where the coefficient θ is the elasticity of demand (this is the loglinearisation of a CES production function relating intermediate outputs to aggregate output):

 $^{^{7}}N$ can be infinite.

⁸The model here differs from the one in Dixon and Kara(2008) in that Dixon and Kara(2008) assume that wages are sticky while goods prices are flexible, whereas here I assume that wages are flexible while goods prices are sticky. This difference does not affect the equilibrium conditions, as the assumption that each household h is twinned with firm f. Thus, in Dixon and Kara(2008), a firm and a household can be thought of as the same entity. Herein, I assume that wages are flexible while goods prices are sticky since the other models are defined in terms of price-setting.

⁹Appendix A provide a detailed discussion of the underlying assumptions of the model and the derivation of the structural equations and therefore, the presentation here is kept brief. See also Dixon and Kara (2007) for a more detailed discussion.

$$y_{it} = \theta(p_t - p_{it}) + y_t \tag{1}$$

Sectoral price levels are given by the average price set in the sector, and the price is averaged over the i cohorts in sector i:

$$p_{it} = \frac{1}{i} \sum_{j=1}^{i} p_{ijt} \tag{2}$$

The log-linearised aggregate price index in the economy is the average of all sectoral prices:

$$p_t = \sum_{i=1}^N \alpha_i p_{it} \tag{3}$$

The inflation rate is given by $\pi_t = p_t - p_{t-1}$.

The Euler condition (20) from the representative household's consumption is given by

$$\tilde{y}_t = E_t \tilde{y}_{t+1} - \eta_{cc}^{-1} (r_t - E_t \pi_{ft+1} - r r_t^N)$$
(4)

where $\tilde{y}_t = y_t - y_t^N$ is the gap between actual output, y_t and the natural level of output, y_t^N . r_t is the nominal interest rate. $rr_t^N = r_t^N - E_t \pi_{t+1}^N = E_t y_{t+1}^N - y_t^N$ denotes the real interest rate when prices are flexible. and the tax rate(τ_t) is at its average level ($\bar{\tau}$). r_t^N , π_{ft}^N and y_t^N denote the nominal interest rate, the inflation rate and the output level when prices are flexible and the tax rate (τ_t) is at its average level ($\bar{\tau}$), respectively.

The natural level of output (i.e., the level of output when prices are flexible and the tax rate (τ_t) is at its average level $(\bar{\tau})$) is given by (derived in the Appendix).

$$y_t^N = \frac{1 + \eta_{_{LL}}}{\eta_{_{cc}} + \eta_{_{LL}}} z_t \tag{5}$$

where $z_t = \log Z_t$ is a productivity shock.

Finally, the productivity shocks $(z_t = \log Z_t)$ follow an AR(1) process. In particular,

$$z_t = \rho_z z_{t-1} + \varepsilon_{zt} \tag{6}$$

where ε_{zt} is an $idd(0, \sigma_z^2)$.

2.3 Price-Setting Rules

Before defining the optimal price setting rules under different models, let us define the optimal price that would occur if prices were perfectly flexible ("the optimal flexible price"). The log-linearised version of the optimal flexible price in each sector¹⁰ is given by

$$p_t^* = p_t + \gamma \tilde{y}_t + \frac{\tilde{\tau}_t}{(1 + \theta \eta_{LL})} \tag{7}$$

with the coefficient on output γ being:

$$\gamma = \frac{\eta_{LL} + \eta_{cc}}{1 + \theta \eta_{LL}} \tag{8}$$

Where $\eta_{cc} = \frac{-U_{cc}C}{U_c}$ is the parameter governing risk aversion, $\eta_{_{LL}} = \frac{-V_{LL}H}{V_L}$ is the inverse of the labour elasticity, θ is the sectoral elasticity and the tax shocks ($\tilde{\tau}_t = \log(1 - \tau_t)$ follow an AR(1) process: $\tilde{\tau}_t = \rho_\tau \tilde{\tau}_{t-1} + \varepsilon_{\tau t}$.

We can represent the alternative price-setting behaviour in terms of two general equations: one for the reset price in sector $i(x_{it})$ and one for the average price in sector $i(p_{it})$. For the GTE, these are¹¹:

$$x_{it} = \sum_{j=1}^{i} \lambda_{ij} E_t p_{t+j-1}^* - a \sum_{j=1}^{i} \sum_{k=j}^{i} \lambda_{ij+k} \pi_{t+j-1}$$
(9)

$$p_{it} = \sum_{j=1}^{i} \lambda_{ij} \left(x_{it-j-1} + a \sum_{k=0}^{j-2} \pi_{t+k-1} \right)$$
(10)

where $\lambda_{ij} = \frac{1}{i}$ and $0 < a \leq 1$ measures the degree of indexation to the past inflation rate. Without indexation (a = 0), the reset price (9) in sector *i* is simply the average (expected) optimal price over the contract length (the

¹⁰Note that the optimal flexible price in each sector is the same. This is because it is based on the demand relation (1) which has the same two aggregate variables $\{p_t, y_t\}$ for each sector. Also, the shocks that hit the sectors are the same.

¹¹I set discount factor (β) to 1. While this assumption simplifies the expositions, the results do not change significantly if I assume $\beta = 0.99$, which is the common assumption in the literature.

nominal price is constant over the contract length). Note that the reset prices will, in general, differ across sectors, since they take the average over a different time horizon. With indexation, the initial price at the start of the contract is adjusted to take into account future indexation over the lifetime of the contract. The average price in sector i (10) is related to the past reset prices and how far they have been indexed.

The two equations (9 and 10) can also represent the simple Calvo economy. To obtain the simple Calvo economy from (9), all reset prices at time t are the same $(x_{it} = x_t)$, the summation is made with $i = \infty$ and $\lambda_{ij} = \omega(1-\omega)^{j-1}$: $j = 1...\infty$. and there is no indexation a = 0. Assuming $0 < a \leq 1$ extends these model to the case in which the prices are indexed to past inflation. The standard equation for the average price is obtained by setting $x_{it} = x_t$, and setting the summation as $i = \infty$ in (10).

In a *GFE*, the trajectory of prices is set at the outset of the contract. Suppose an i- period contract starts at time t; then the sequence of prices chosen from t to t + i - 1 is $\{E_t p_{t+s}^*\}_{s=0}^{s=i-1}$. Hence, the average price in sector i at time t is

$$p_{it} = \sum_{j=1}^{i} \lambda_{ij} E_{t-j+1} p_t^*$$
(11)

which is the average of the best guesses of each cohort for the optimal flexible price to be holding at t and embodies the "sticky information" idea in Fischer contracts: part of current prices are based on old information. In the *GFE*, since cohorts are of equal size within sector i, $\lambda_{ij} = \frac{1}{i}$. The Mankiw-Reis sticky-information (*SI*) model has $\lambda_{ij} = \omega (1 - \omega)^{j-1}$: $j = 1..\infty$.

2.4 Monetary Policy Rules

I assume that the central bank follows a simple Taylor-type rule under which the interest rate reacts to the lagged interest rate, inflation and the output gap.

$$r_t = \phi_r r_{t-1} + \phi_\pi \pi_t + \phi_y \tilde{y}_t \tag{12}$$

The ϕ -coefficients in front of the targeting variables are chosen to minimise welfare loss.

To provide a measure of the relative performance of this policy rule in a given model, I also report its relative loss, which gives the ratio between the loss under the rule and the first best welfare level obtainable in that model. The welfare level under the first best can be obtained by using Lagrangian methods. More specifically, this is the welfare level that can be obtained by maximising the welfare function subject to the equilibrium conditions. I obtain the first order conditions of this problem by differentiating the Lagrangian with respect to each of the endogenous variables and setting these conditions to zero. I then combine the first order conditions together with equilibrium conditions and calculate the implied welfare level¹².

3 Welfare Functions: Woodford's Approximation

In this section, a utility-based objection function is derived to provide a benchmark for evaluating the performance of alternative monetary policy rules. We can represent the welfare function for each model in terms of a general equation. For the GTE, the welfare function is given by (derived in Appendix B)

$$W_t = -\frac{U_c(C)C}{2}L_t + t.i.p$$
(13)

where C is the steady state consumption, $U_c(C)$ is the marginal utility of consumption and *t.i.p* collects all the terms that are independent of policy. The loss function, L_t , is given by

$$L_{t} = (\eta_{cc} + \eta_{LL})\tilde{y}_{t}^{2} + \theta (1 + \eta_{LL}\theta) \sum_{i=1}^{N} \sum_{j=1}^{i} \alpha_{i}\lambda_{ij} (p_{ijt} - p_{t})^{2}$$
(14)

where $\lambda_{ij} = \frac{1}{i}$. This expression implies that welfare loss depends on the variance of the output gap and on the cross-sectional price dispersion. When there is only one type of contract length in the economy, the function reduces to the welfare function in a standard one sector Taylor model.

The loss function reduces to the loss function in the Calvo model as in Woodford (2003, p. 396), when all reset prices at time t are the same $(p_{ijt} = p_{jt})$, the summation is made with $i = \infty$ and $\lambda_{ij} = \omega(1-\omega)^{j-1}$: $j = 1...\infty$ and there is no indexation. Woodford (2003) shows in the Calvo model and in its variant with indexation that, the welfare costs of cross-sectional price

 $^{^{12}\}mathrm{I}$ use the Dynare's "olr" to perform these calculations

dispersion can be summarised in terms of variability of current and lagged inflation rates¹³. Thus, the loss function can be rewritten as

$$L_{t} = \left[(\eta_{cc} + \eta_{LL}) \tilde{y}_{t}^{2} + \frac{\omega^{2}}{1 - \omega} (1 + \eta_{LL}\theta) (\pi_{t} - a\pi_{t-1}) \right]$$
(15)

The loss function gives the loss function in the sticky information model, as in Ball, Mankiw and Reis (2005) (p. 13), when $\lambda_{ij} = \omega (1-\omega)^{j-1}$: $j = 1..\infty$, $p_{ijt} = p_{jt}$, and $p_{jt} = E_{t-j+1}p_t^* - p_t$, When there is only one type of contract length in the economy $(\lambda_{ij} = \frac{1}{i})$, the function reduces to the welfare function in a standard Fischer model. Ball et al. (2005) show that the crosssectional price variability in the *SI* can be expressed in terms of aggregate variables

$$\sum_{j=1}^{i} \lambda_j (p_{jt} - p_t) = \sum_{j=1}^{i} \eta_j (p_t - E_{t-j} p_t)$$
(16)

where $\eta_j = \frac{\omega(1-\omega)^{j-1}}{(1-(1-\omega)^j)(1-(1-\omega)^{j+1})}.$

4 The Choice of Parameters

I begin with a calibration in a GTE. I consider a special GTE: Calvo-GTE, in which the share of each duration across firms $(\alpha_i = \omega^2 i (1 - \omega)^{i-1} : i = 1...\infty)$ is the same as generated by the Calvo model. The discussions in Bils and Klenow (2004) and Nakamura and Steinsson (2007a) suggest that the Calvo distribution is not a bad approximation of empirical distribution. The two key parameters in this model are ω and γ . ω is the parameter that indicates the degree of nominal rigidity in the economy. $\gamma = \frac{\eta_{cc} + \eta_{LL}}{1 + \eta_{LL}\theta}$ is the parameter on the output gap in the price setting equation for each sector. Following the literature (e.g. Walsh (2005), Woodford (2003)), I set $\theta = 7.88$, $\eta_{cc} = 1$, $\eta_{LL} = 1.2$. The implied value of $\gamma = 0.2$. Recent work of Nakamura and Steinsson (2007a) suggests $\omega = 0.25^{14}$. There is no indexation (a = 0).

¹³In the GTE, this is not the case: the welfare costs of cross-sectional dispersion cannot be summarised in terms of variability of inflation and must be given explicitly in terms of variances of relative prices. This is mainly because in the GTE, there is a distribution of sector-specific reset prices in each period.

¹⁴Note that this number excludes sales and substitution-related prices changes. The Bils and Klenow (2004) dataset indicate a lower degree of nominal rigidity: $\omega = 0.4$ (see

The other models are calibrated according to the macro-estimates. In the *IC* model, the key parameters are a, ω and γ . A range of estimates for a and ω for the U.S. are reported in Table 1. The estimates of a indicate that the degree of indexation is around 0.66 - 0.84. The estimates of ω indicate that the proportion of firms that set prices is between 0.07 - 0.17. Given these numbers, I follow Rabanal and Rubio-Ramirez (2005) and set a = 0.76 and $\omega = 0.17^{15}$. In the Calvo model, I also follow Rabanal and Rubio-Ramirez (2005) and set $\omega = 0.22$. In both models, as in the case of the Calvo-*GTE*, I set $\gamma = 0.2$.

To calibrate the SI model, I choose among the values estimated by Coibion (2008) and Mankiw and Reis (2007). The key parameters in this model are ω and γ . Coibion (2008) argues that low values of ω and γ are necessary for the model to match the persistence in the data. More specifically, he finds that the model comes closer to matching the data when $\gamma \approx 0.03$ and $\omega \approx 0.7$ (see Coibion(2008, p. 28, Figure 3)). Findings reported in Mankiw and Reis(2007, p. 610, Table 1) and in Kiley (2006, p.112, Table 3) indicate the same conclusion. Mankiw and Reis estimate that $\theta = 34.1$ and $\omega = 0.7^{16}$. The value of $\theta = 34.1$, along with $\eta_{cc} = 1$ and $\eta_{LL} = 1.2$, implies

¹⁶These values are based on maximum-likelihood estimates. The authors also estimate their model using Bayesian methods. In this case, the values are slightly lower: $\theta = 20.5$ and $\omega = 0.657$. Here I set the parameters at the maximum-likelihood estimates since Mankiw and Reis themselves use these estimates when reporting the impulse response function of inflation to monetary policy. In any case, using the Bayesian estimates rather than the maximum-likelihood estimates does not change results significantly.

Dixon and Kara (2008)). This is because Bils and Klenow (2004) include prices changes due to sales and substitutions. In any case, calibrating the Calvo-GTE by using $\omega = 0.4$ does not affect the conclusions significantly.

¹⁵Note that recent work by Levin, Onatski, Williams and Williams (2005) almost argues that the Calvo model without indexation matches the US data well. Specifically, it is argued that the degree of indexation in the US is as low as 13%. The Levin et al. (2005) conclusion is surprising because Levin et al. (2005) use a model that is very similar, if not identical, to that in Del-Negro, Schorheide, Smets and Wouters (2007). Unfortunately, there is no hint to be found in Levin et al. (2005) as to why the degree of indexation is substantially lower in their model. Searching for the reason would lead me beyond the scope of purpose of this paper. Therefore, here I take the Del-Negro et al. (2007) view, which reflects the common view. Recent work by Smets and Wouters (2007) replace the Dixit-Stiglitz aggregator with a more complicated aggregator (i.e. a Kimball aggregator (Kimball (1995))) and find that doing so reduces the degree of indexation. The degree of indexation in this model is 0.24. However, such an assumption has significant implications for optimal policy design. I leave this issue for a separate paper and here stick to the standard Dixit-Stiglitz aggregator.

that $\gamma = 0.05$. Thus, I follow Mankiw and Reis (2007) and set $\theta = 34.1$ and $\omega = 0.7^{17}$. There is no indexation. I will also report results with a lower value of ω and a higher value of γ .

Finally, I assume that the shocks processes are the same in all models. The productivity shocks follow an AR(1) process. The serial correlation parameter is assumed to be $\rho_z = 0.95$, and the standard deviation of the shock is set to be $\varepsilon_{zt} = 0.007$. These are standard assumptions in the real business cycle literature. For the tax shocks, following Walsh (2005), the serial correlation parameter is calibrated as $\rho_{\tau} = 0.80$ and the standard deviation of the shock is set to be $\varepsilon_{\tau t} = 0.024$. Walsh obtains these values by estimating an AR(1) process for detrended log fiscal variables, using the dataset on tax revenues complied by Blanchard and Perotti (2002).

5 Inflation Dynamics and Micro-evidence on Prices

One common way to assess a model's empirical validity is to see whether it can generate plausible responses to monetary policy shocks (Woodford (2003) and Mankiw and Reis (2002)). It is widely agreed that inflation exhibits a delayed response to monetary policy. That is, the maximum effect of a policy occurs sometime after the policy: that is, there is a hump-shaped response (Christiano et al. (2005)). To study the response of inflation to monetary policy in each model, I assume that the central bank follows a Taylor rule, under which the nominal interest rate responds to inflation and the output gap. Specifically, I consider a Taylor rule of the following form

$$r_t = \phi_r r_{t-1} + \phi_\pi \pi_t + \phi_y \tilde{y}_t - \xi_{rt} \tag{17}$$

where $\xi_{rt} = \rho_r \xi_{rt-1} + \varepsilon_{rt}$. The literature suggests that a simple rule provides a reasonable description of US monetary policy. Following Rudebusch (2002), I set $\phi_r = 0$, $\phi_{\pi} = 1.24$, $\phi_y = 0.33$ and $\rho_r = 0.92^{18}$. Rudebusch (2002) argues that interest rate inertia may reflect responses to serially-correlated shocks rather than a desire to gradually change interest rates.

 $^{^{17}}$ Note, however, that $\theta=34.1$ is implausibly high. $\,\theta$ is typically calibrated between 6 and 10.

 $^{^{18}}$ The same rule is used in Mankiw and Reis(2007).

Under the calibrated parameters, I compute the impulse responses of inflation for the IC and SI following a positive monetary policy shock¹⁹. I then compare the responses of inflation response in the Calvo-GTE. Figure 1a and Figure 1b display the inflation responses for the IC and SI, respectively. As is evident from these figures, the models fit the empirical evidence regarding how inflation responds to monetary policy shocks. In response to a monetary expansion in each model, inflation rises gradually, following the hump-shaped pattern found in empirical studies. For the sake of comparison, Figure 1a also plots the inflation response in the Calvo model. As is evident, the Calvo model cannot capture inflation dynamics: inflation always peaks on impact and is less persistent as compared to other models. Finally, Figure1c reports the inflation response in the Calvo-GTE. As the figure shows, the inflation response in the Calvo-GTE is very similar to that in the IC and the SI: inflation rises gradually and then follows a hump-shaped pattern.

Given these findings, it is safe to conclude that there is no material difference between the inflation response in the Calvo-GTE versus in the ICand SI models. However, a difference between the GTE and its popular alternatives arises when it comes to the micro-data. Namely, the GTE is consistent with the micro-evidence, whereas the IC and SI are not.

5.1 Discussion

It is important to note that the size of the inflation responses in the Calvo-GTE and the IC are very similar. However, inflation response in the SI is larger than those in the other models. Kiley(2006) finds that the IC model performs empirically better than the SI. This difference in responses seems to explain Kiley's (2006) finding. The reason why inflation response is larger in the SI model is easy to understand; as noted in Dixon and Kara(2008), in this model, the length of the contract does not affect price setting behaviour, whereas in the other two models it does. In the SI, firms set their prices according to the optimal flexible price, subject to information constraints. As is well known, the staggered contract structure dampens the effect of a shock on prices.. Since in the SI model, the contract length does not influence price setting behaviour, the response of inflation is much larger than in the other models. If I were to assume that ω is lower, at, say $\omega = 0.25$, as in the

¹⁹All calculations are performed by using Dynare (see Juillard (1996))

Calvo-GTE, Figure 2 shows that, the scale of the response in the SI would be more in line with the scales in the IC and Calvo-GTE. The reason for this is also easy to understand. The optimal flexible price depends mainly on the general price level, which includes the prices of firms that set their prices with out of date information (i.e. uninformed firms). A lower ω indicates that the share of uninformed firms in the economy is higher. This implies that the aggregate price level will react sluggishly. This means that even the firms that set their prices with full information will not change their prices as much as they otherwise would. However, as Figure 2 further shows, with a lower ω , the model generates too much persistence²⁰. This finding is line with the findings reported in Coibion(2008).

Mankiw and Reis (2002) argue that the key empirical fact that is hard to match is the delayed response of inflation. If one takes the Mankiw and Reis view, as this paper does, then the conclusion would be that all models considered in this paper are very similar.

6 Results

It is obvious from previous section that there is no material difference between the Calvo-GTE approach and its popular alternatives in terms of inflation persistence. A difference arises when it comes to micro-data: the GTE is consistent with respect to micro-data, whereas the IC and the SI are not.

Having established these facts, I can now use the Calvo-GTE to answer the following question: what are the consequences of implementing a policy rule that is optimal from the perspective of the IC and SI models if the true model is the GTE?

To answer this question, I first consider the case in which the central bank uses the IC model when formulating its monetary policy. More specifically, the central bank simulates the IC model to find the optimal reaction coefficients in the policy rule. I then compute the welfare loss under such a policy in the true economy (i.e. the Calvo-GTE). I then repeat the same experiment for the case in which the central bank uses the SI model when formulating its policy instead of the IC model. Before carrying out this experiment, it is essential to establish that the optimised three- parameter rule performs well in each model. I will also discuss how policy conclusions are affected by whether inflation inertia arises in a manner consistent with the

²⁰Assuming $\gamma = 0.2$, instead of $\gamma = 0.05$, does not significantly change these conclusions.

micro-data. Welfare levels (W) are expressed in terms of the equivalent percentage decline in terms of steady state consumption, which can be obtained by dividing W by U_cC . Welfare levels under optimal policy corresponds to those discussed in section 2.4.

6.1 Optimal Policy and Micro-evidence on Prices

In this section, I evaluate the performance of the optimised rule for each model. Table 2 displays the welfare losses under such a rule for each model when coefficients are chosen optimally. The losses reported in the table are expressed as ratios of the welfare losses in each model to the loss under the optimal monetary policy. As is evident from the table, the optimised rule performs reasonably well in each model: the relative loss in each model is less than $10\%^{21}$.

However, as Table 3 also shows, the models differ in their recommendations for the optimal policy. Reported are the optimal ϕ -coefficients that minimise the welfare loss in each model. A key difference arises when it comes to how aggressive the central bank should be in its response to inflation. According to the SI model, the central bank should respond aggressively The coefficient of inflation in the policy rule is as high as 11. to inflation. This finding is in line with the findings reported in Reis (2008). Reis (2008) studies optimal monetary policy by using the sticky information model and finds that the optimal value of ϕ_{π} is larger than the typical estimates of this parameter. For example, as noted earlier, Rudebusch estimates $\phi_{\pi} = 1.24$. Therefore, Reis (2008) concludes that "interest rates should respond more aggressively to inflation than they have". However, if one employs the ICmodel for the monetary policy analysis, the conclusion would be exactly the opposite; interest rates should respond less aggressively to inflation than they have. According to the *IC* model, the coefficient is much less than that suggested by the SI model and is around 0.6. The Calvo-GTE suggests a value of $\phi_{\pi} = 1.1$.

In tracing the source of the reasons for differences in policy recommendations, let me begin by considering the mechanism at work in the Calvo-GTEmodel. To understand the mechanism at work in the GTE, first note that price stickiness dampens the effect of a shock on prices. In other words,

 $^{^{21}}$ This level of relative loss has been considered to be reasonable by previous studies (see, for example, Levin and Williams (2003) and Huang and Liu (2005))

when prices are sticky, firms cannot change their prices as much as they would when prices are flexible. This sluggish adjustment means that a small change in prices implies a large change in the optimal prices of firms. This is because when firms reset their prices, they take into account the fact that they will have to charge the same price throughout the lenght of the contract. Since there is a trade-off between price stability and output gap stability, the large movements in firm prices require large movements in the output gap to control price stability. Therefore, a policy that reacts inflation strongly is costly. As a consequences, the coefficient of inflation in the policy rule is not large, with a value around 1.

Now consider the IC model. The finding that the central bank should not react to inflation strongly in this model is in line with the common finding in the literature that the best policy response based on the Calvo model with indexation is to let inflation adjust sluggishly (see, for example, Woodford (2003, p. 482-83)). This is due to the fact that the assumption of indexation alters the loss function of the central bank in an important way: the central bank aims to stabilise fluctuations in $(\pi_t - a\pi_{t-1})$, rather than π_t . To put it differently, stabilising fluctuations in inflation do not improve welfare, as the central bank cares about the fluctuations in $(\pi_t - a\pi_{t-1})$. On the contrary, given the policy trade-off, this policy would be very costly, as it would require larger output gap fluctuations. Table 3 confirms this intuition. Reported in the table are standard deviations of \tilde{x}_t , π_t and $\pi_t - a\pi_{t-1}$. Under the optimal Taylor rule, π_t is more volatile than $\pi_t - a\pi_{t-1}$. However, the volatility in π_t is irrelevant for welfare loss. For example, if the central bank reacts strongly to inflation, say $\phi_{\pi} = 1.1$, which is the value suggested by the Calvo-GTE model and holds other factors constant, π_t becomes less volatile; however, this leads to greater output variability and, therefore, higher welfare loss. Therefore, when the IC model is employed for monetary policy analysis, the conclusion is that it is costly to stabilise the fluctuations in inflation.

Finally, we can look at the SI. The question here is why the SI model favours a policy that reacts to inflation strongly. The main reason is that the policy trade-off that the central bank faces is less important than the policy trade-offs in the other models. That is, in this model reacting to inflation strongly requires smaller output gap movements to control inflation. The reason for this is an aspect of the model that is inconsistent with the microdata: a different price can be chosen within the span of contract. In this model, the length of the contract does not affect the price setting behaviour: regardless of the contract length, for any period t, the price set by a given firm will be its best guess at what the optimal flexible price will be in that period. In sharp contrast, in the GTE, the contract length does affect pricesetting behaviour, as is the case in the Calvo model. As discussed in section 2.3, in the GTE, the reset price in sector i is the average optimal flexible price over the length of the contract. The nominal price is constant over the contract length. In the Calvo model, if ω is large, for example, then the firm gives more weight to the near future, since a large ω means that the price it sets now is less likely to survive, and vice versa. In the SI, prices also adjust sluggishly due to the fact that some firms adjust their prices while using out of date information. However, it appears that in a model in which price stickings arises due to information rigidity, it is less costly to react to inflation strongly than in a model in which price stickiness arises due to staggered contracts. This can be demostrated by choosing a value of ω for the SI model that brings the model's predictions closer to those in the Calvo-GTE. All of the other parameters in the SI are calibrated as in the Calvo-GTE. A value of ω that brings the SI's predictions closer to those in the Calvo-GTE is $\omega = 0.1$, which is lower than the value of ω in the Calvo-GTE. As Table 5 shows, when $\omega = 0.1$, the policy recommendations of the SI model are very similar to those of the Calvo-GTE. The relative losses in the two models are almost exactly the same and the optimal policy parameters are also very similar. It should also be noted that the value of $\omega = 0.1$ is much lower than what the model would require to explain the inflation persistence we observe in the data. This analysis thus allows me to conclude that the policy trade-off in the SI model is less severe and, therefore, it is less costly to react to inflation strongly.

The analysis above indicates that the features of the IC and SI models that are inconsistent with the micro-data on prices affect the degree of trade-off between inflation and output gap stabilization: the policy trade-off is severe in the IC model, whereas in the SI model it is not. To put it differently, the SI model underestimates the cost of output gap fluctuations, unless a sufficiently low value of ω is assumed, whereas the Calvo-GTE overestimates the cost. The importance of this point for optimal policy design can also be demonstrated by considering a policy that reacts to inflation and the lagged interest rate only (i.e., an extreme inflation-targeting policy). As discussed above, given the policy trade-off, such a policy would typically require larger output gap movements and, therefore, lead to larger welfare losses. This turns out not to be the case according to the SI. As the table shows, under the benchmark calibration, this policy does not visibly affect the welfare loss unless a substantially lower value of ω is assumed. Perhaps not surprisingly, the *IC* yields a very different conclusion: ignoring fluctuations in the output gap is very costly. The relative loss increases from 9% to 26%. The Calvo-*GTE* suggests that such a policy leads to larger welfare losses. However, the cost is not as high as what the *IC* model suggests. The relative loss increases from 7% to 9%. The more general conclusion from comparing the results in Tables 2 and 5 is that the *IC* and *SI* models can potentially provide a misleading assessment of outcomes under alternative policy rules.

Two conclusions emerge from this discussion. First, an optimised Taylor rule performs reasonably well across all models. Second, policy conclusions that arise from the SI and IC are significantly affected by the aspects of these models that are inconsistent with the micro-data on prices.

6.2 Robustness to the Calvo - GTE

I have shown that the optimised Taylor rule performs reasonably well across all models under consideration here and I have demonstrated that the aspects of the IC and SI that are inconsistent with the micro-evidence on prices affect policy conclusions. I now turn to my main question: how much welfare loss would be incurred if a central bank employed a model for monetary policy analysis that was inconsistent with the micro-evidence on prices?

I assume that the GTE is the true economy. I begin by considering the case in which the central bank formulates its policy by employing the IC model. This case is of particular interest, since the IC model guides much of the monetary policy discussion. Column 2 of Table 6 reports the welfare loss in the GTE under the rule that is optimal from the perspective of the IC model. As is evident from Table 6, employing a rule that is optimal from the perspective of the IC model can lead to a poor outcome in the Calvo-GTE. More specifically, the relative welfare loss in the Calvo-GTE is as high as 30%. Note that a higher degree of indexation can lead to even larger welfare losses. As a point of reference, column 4 of Table 6 repeats the same experiment as in Table 6 column 2 but assumes full indexation. That is a = 1, as in Christiano et al. (2005). Evidently, formulating monetary policy by assuming a = 1, can lead to a disastrous outcome in the Calvo-GTE: the relative welfare loss is as high as 137%.

Turning to the SI, Column 3 Table 6 shows the results of the same exercise as in Column 2 of Table 6 but assumes that the policy generating

model is the SI. The results reported there indicate that the conclusion in the previous experiment carries over to a setting in which firms adjust their prices according to the Sticky-information model. When the SI is assumed, the relative loss is around 30%.according to in the Calvo-GTE

The conclusion, therefore, is that a policy that is optimal based on a model that is inconsistent with the micro-data can lead to significant welfare loss in an economy that is consistent with micro-evidence. This is the case even though all three approaches generate almost exactly the same degree of inflation persistence.

In light of the discussion in the previous section, the source of the large welfare losses in the Calvo-GTE is easy to identify. The SI recommends a larger value for ϕ_{π} . Such a policy would stabilise the fluctuations in inflation in the Calvo-GTE at the cost of significantly greater output gap variability. In contrast, the IC recommends a small value for ϕ_{π} . The resulting policy would not be sufficient to stabilise inflation in the Calvo-GTE.

7 Summary and Conclusions

The failure of the Calvo model to account for key macro-evidence has led to two main responses in the literature: the introduction of indexation to the Calvo model (IC) and the adoption of Fischer contracts and a Calvo distribution of contract lengths (Sticky Information) (SI). Both of these theories are inconsistent with the micro-data on prices, as all prices change at each period.

The purpose of this paper has been to investigate the consequences of employing models that are inconsistent with the micro-data for optimal monetary policy design. To do so, I used a Generalized Taylor Economy (GTE). The GTE is a generalisation of the simple Taylor model and explicitly allows for sectoral heterogeneity with the ranges of contract lengths suggested by the data. I consider a special case of the GTE, namely, the Calvo-GTE, in which the distribution of contract lengths is generated by the Calvo model, as in the IC and SI. The Calvo-GTE is consistent with the micro-data and can model inflation dynamics as well as its popular alternatives.

The findings reported in this paper suggest that policy conclusions are significantly affected by the aspects of the IC and SI models that are inconsistent with the micro-data on prices. More specifically, the aspects of the models that are inconsistent with these micro-data affect the degree of

policy trade-off between price stability and output gap stability. Therefore, a failure to recognise the importance of micro-data can lead to misleading analysis regarding the policy trade-offs that policymakers face and may result in the design of policy rules that may not be appropriate for implementation. Indeed, policies that are optimal from the perspective of these models can lead to significant welfare losses in the Calvo-GTE.

I conclude, therefore, that the micro-data on prices are important for optimal monetary policy design. That is, it is a mistake to place undue emphasis on the macro-data at the expense of micro-data, as has been the case with many previous studies on this topic.

References

- Ball, L., Mankiw, N. G. and Reis, R.: 2005, Monetary policy for inattentive economies, *Journal of Monetary Economics* **52**(4), 703–725.
- Bils, M. and Klenow, P.: 2004, Some evidence on the importance of sticky prices, *Journal of Political Economy* 112(5), 947–985.
- Blanchard, O. and Perotti, R.: 2002, An empirical characterization of the dynamic effects of changes in government spending and taxes on output, *The Quarterly Journal of Economics* **117**(4), 1329–1368.
- Chari, V. V., Kehoe, P. J. and McGrattan, E. R.: 2008, New Keynesian models: Not yet useful for policy analysis, *Federal Reserve Bank of Minneapolis Staff Report 409*.
- Christiano, L. J., Eichenbaum, M. and Evans, C. L.: 2005, Nominal rigidities and the dynamic effects of a shock to monetary policy, *Journal of Political Economy* 113(1), 1–45.
- Christiano, L., Motto, R. and Rostagno, M.: 2008, Financial factors in business cycles, *mimeo*. Available from: http://faculty.wcas.northwestern. edu/~lchrist/research/ECB/manuscript.pdf (accessed November 23, 2008).
- Christoffel, K., Coenen, G. and Warne, A.: 2008, The new area-wide model of the euro area: a micro-founded open-economy model for forecasting and policy analysis, *Working paper series 944, European Central Bank*
- Coibion, O.: 2008, Testing the sticky information Phillips curve, *The Review* of Economics and Statistics (forthcoming).
- Del-Negro, M., Schorheide, F., Smets, F. and Wouters, R.: 2007, On the fit and forecasting performance of new Keynesian models, *Journal of Business and Economic Statistics* 25(2), 123–162.
- Dhyne, E., Alvarez, L. J., Bihan, H. L., Veronese, G., Dias, D., Hoffmann, J., Jonker, N., Lunnemann, P., Rumler, F. and Vilmunen, J.: 2005, Price setting in the euro area: some stylized facts from individual consumer price data, *Working paper series 524, European Central Bank*.

- Dixon, H. and Kara, E.: 2006a, How to compare Taylor and Calvo contracts: a comment on Michael Kiley, *Journal of Money, Credit and Banking* **38**(2).
- Dixon, H. and Kara, E.: 2007, Persistence and nominal inertia in a Generalized Taylor Economy: how longer contracts dominate shorter contracts?, Working paper series 2007/1, Cardiff Business School.
- Dixon, H. and Kara, E.: 2008, Can we explain inflation persistence in a way that is consistent with the micro-evidence on nominal rigidity?, *Cardiff Economics Working Paper 2008/22*.
- Edge, R.: 2002, The equivalence of wage and price staggering in monetary business cycle models, *Review of Economic Dynamics* 5, 559–585.
- Fischer, S.: 1977, Long-term contracts, rational expectations, and the optimal money supply rule, *Journal of Political Economy* 85(1), 191–205.
- Huang, K. X. and Liu, Z.: 2005, Inflation targeting: What inflation rate to target?, *Journal of Monetary Economics* **52**(8), 1435–1462.
- Juillard, M.: 1996, Dynare: A program for the resolution and simulation of dynamic models with forward variables through the use of a relaxation algorithm, CEPREMAP, Working Paper No. 9602. (http://www.cepremap.cnrs.fr/dynare).
- Justiniano, A. and Primiceri, G. E.: 2008, The time varying volatility of macroeconomic fluctuations, *American Economic Review* **98**, 604–641.
- Kimball, M.: 1995, The quantitative analytics of the basic neomonetarist model, Journal of Money, Credit and Banking 27(4), 1241–1277.
- Levin, A. T., Onatski, A., Williams, J. C. and Williams, N.: 2005, Monetary policy under uncertainty in micro-founded macroeconometric models, *NBER Working Paper No.* 11523.
- Levin, A. T. and Williams, J. C.: 2003, Robust monetary policy with competing reference models, *Journal of Monetary Economics* **50**(5), 945–975.
- Mankiw, N. G. and Reis, R.: 2002, Sticky information versus sticky prices: A proposal to replace the new keynesian phillips curve, *The Quarterly Journal of Economics* 117(4), 1295–1328.

- Mankiw, N. G. and Reis, R.: 2007, Sticky information in general equilibrium, Journal of the European Economic Association 5(2-3), 603–613.
- Nakamura, E. and Steinsson, J.: 2007a, Five facts about prices: A reevaluation of menu cost models, *Quarterly Journal of Economics* (forthcoming).
- Rabanal, P. and Rubio-Ramirez, J. F.: 2005, Comparing new Keynesian models of the business cycle, *Journal of Monetary Economics* 52, 1151– 1166.
- Reis, R.: 2008, A Sticky-Information General-Equilibrium Model for Policy Analysis. In: Monetary Policy under Uncertainty and Learning, edited by K. Schmidt-Heubel and C. Walsh, Central Bank of Chile, number forthcoming.
- Rudebusch, G.: 2002, Term structure evidence on interest rate smooting and monetary policy interia, *Journal of Monetary Economics* 49(6), 1161– 1187.
- Schorfheide, F.: 2008, DSGE model-based estimation of the new Keynesian Phillips curve, *mimeo*.
- Smets, F.: 2008, DSGE modelling at the european central bank, mimeo . Available from: http://www.bcb.gov.br/Pec/seminarios/ SemMetInfl2008_2/Apresenta%C3%A7%C3%A30%20Frank%Smets. pdf (accessed November 23, 2008).
- Smets, F. and Wouters, R.: 2003, An estimated dynamic stochastic general equilibrium model of the euro area, *Journal of the European Economic* Association 1(5), 1123–1175.
- Smets, F. and Wouters, R.: 2005, Comparing shocks and frictions in US and Euro Area business cycles: A bayesian DSGE approach, *Journal of Applied Econometrics* 20, 161–183.
- Smets, F. and Wouters, R.: 2007, Shocks and frictions in US business cycles: A bayesian DSGE approach, American Economic Review 97(32), 586– 607.

- Walsh, C.: 2005, Endogenous objectives and the evaluation of targeting rules for monetary policy, *Journal of Monetary Economics* 52, 889–911.
- Woodford, M.: 2003, Interest and prices: Foundations of a theory of monetary policy, *Princeton University Press, Princeton, NJ*.
- Woodford, M.: 2007, Interpreting inflation persistence, *Journal of Money*, Credit and Banking **39**, 203–210.

Appendix:

A The Model

A.1 Households

The representative household h has a utility function given by

$$U_{h} = E_{t} \left[\sum_{t=0}^{\infty} \beta^{t} \left[U(C_{ht}) + V(1 - H_{ht}) \right] \right]$$
(18)

where C_{ht} , H_{ht} are household h's consumption and hours worked respectively, t is an index for time, $0 < \beta \leq 1$ is the discount factor, and $h \in [0, 1]$ is the household specific index.

The household's budget constraint is given by

$$P_t C_{ht} + \sum_{s^{t+1}} Q(s^{t+1} \mid s^t) B_h(s^{t+1}) \le B_{ht} + (1 - \tau_t) W_{ht} H_{ht} + \Pi_{ht} + T_{ht} \quad (19)$$

where $B_h(s^{t+1})$ is a one-period nominal bond that costs $Q(s^{t+1} | s^t)$ at state s^t and pays off one dollar in the next period if s^{t+1} is realized. B_{ht} represents the value of the household's existing claims given the realized state of nature. W_{ht} is the nominal wage, Π_{ht} is the profits distributed by firms and $W_{ht}H_{ht}$ is the labour income. τ_t denotes the labour income tax²². Finally, T_t consists of transfers.

The first order conditions derived from the consumer's problem are as follows:

$$u_{ct} = \beta R_t E_t \left(\frac{P_t}{P_{t+1}} u_{ct+1} \right) \tag{20}$$

$$\sum_{s_{t+1}} Q(s^{t+1} \mid s^t) = \beta E_t \frac{u_{ct+1} P_t}{u_{ct} P_{t+1}} = \frac{1}{R_t}$$
(21)

$$W_{it} = \frac{\theta}{\theta - 1} \frac{1}{(1 - \tau_t)} \frac{V_L (1 - H_{it})}{\frac{u_c(C_t)}{P_t}}$$
(22)

²²Note that the labour tax is a policy variable and I assume that the government sets it equal across consumers. Therefore, τ does not have the subscript h.

Equation (20) is the Euler equation. Equation (21) gives the gross nominal interest rate. Equation (22) shows that the optimal wage. The index h is dropped in equations (20) and (22), which reflects our assumption of complete contingent claims markets for consumption and implies that consumption is identical across all households in every period $(C_{ht} = C_t)$.

A.2 Firms

A typical firm in the economy produces a differentiated good which requires labour as the only input, with a CRS technology represented by

$$Y_{ft} = Z_t L_{ft} \tag{23}$$

 $f \in [0, 1]$ is firm specific index. Differentiated goods Y_{ft} are combined to produce a final consumption good Y_t . The production function here is CES and corresponding unit cost function P_t

$$Y_t = \left[\int_0^1 Y_{ft} \frac{\theta_{-1}}{\theta} df\right]_1^{\frac{\theta}{\theta_{-1}}}, \qquad (24)$$

$$P_t = \left[\int_0^1 P_{ft}^{1-\theta} df \right]^{\frac{1}{1-\theta}}$$
(25)

The demand for the output of firm f is given by

$$Y_{ft} = \left(\frac{P_{ft}}{P_t}\right)^{-\theta} Y_t \tag{26}$$

The firm chooses $\{P_{ft}, Y_{ft}, L_{ft}\}$ to maximize profits subject to (23, 26), yields the following solutions for price, output and employment at the firm level given $\{Y_t, W_{ft}, P_t\}$.

$$P_{ft} = \frac{\theta}{\theta - 1} \frac{W_{ft}}{Z_t} \tag{27}$$

$$Y_{ft} = \left(\frac{\theta}{\theta - 1}\right)^{-\theta} \left(\frac{W_{ft}}{Z_t P_t}\right)^{-\theta} Y_t$$
(28)

$$L_{ft} = \left(\frac{\theta}{\theta - 1}\right)^{-\theta} \left(\frac{W_{ft}}{Z_t P_t}\right)^{-\theta} \left(\frac{Y_t}{Z_t}\right)$$
(29)

Where $\frac{\theta}{\theta-1}$ measures the markup. Price is a markup over marginal cost, which depends on the wage rate (W_{ft}) and productivity shocks. Output and employment depend on the real wage, total output in the economy and productivity shocks.

Using (27), aggregating for firm f in sector i, substituting out for W_{it} in the resulting equation using the optimal labour supply condition (22), using the labour demand function (29) to substitute out for L_{it} and log-linearizing the resulting equation, I obtain the price level when prices are full flexible $(7)^{23}$

$$p_t^* = p_t + \gamma y_t - \frac{(1 + \eta_{LL})}{(1 + \theta \eta_{LL})} z_t + \frac{\tilde{\tau}_t}{(1 + \theta \eta_{LL})}$$
(30)

Note that the optimal flexible price in each sector is the same $(p_{it} = p_{it}^* = p_t^*)$. This is because it is based on the demand relation (1) which has the same two aggregate variables $\{p_t, y_t\}$ for each sector. Also, the shocks that hit the sectors are the same.

Since the optimal flexible price is the same in each sector $p_{it} = p_t$, the output level when prices are fully flexible is given by.

$$y_t^* = \frac{(1+\eta_{LL})}{(\eta_{cc}+\eta_{LL})} z_t + \frac{\tilde{\tau}_t}{(\eta_{cc}+\eta_{LL})}$$
(31)

The natural level of output is obtained when there are no markup shocks:

$$y_t^N = \frac{(1 + \eta_{LL})}{(\eta_{cc} + \eta_{LL})} z_t$$
(32)

This equation implies that the natural level output varies with the pro-

 $^{^{23}{\}rm I}$ follow the notational convention that lower-case symbols represents log-deviations of variables from the steady state.

ductivity shocks.

B Derivation of the welfare function²⁴

A second-order approximation of the period utility $U(C_t)$ around steady state yields:

$$U_t(C) = U_C(C)C(c_t + \frac{1 - \eta_{cc}}{2}c_t^2) + t.i.p,$$
(33)

where c_t denotes the log-deviation of consumption from steady state, t.i.p collects all the terms that are independent of policy and $O(||a||^3)$ summarizes all terms of the third or higher orders.

Using the fact that $c_t = y_t$ in the model and the definition $\hat{y}_t = y_t - \bar{y}_t$, (33) can be expressed in terms of the output gap

$$U_t(C) = U_C C\left(\hat{y}_t + \frac{1 - \eta_{cc}}{2}\hat{y}_t^2 + (1 - \eta_{cc})\hat{y}_t\bar{y}_t\right) + t.i.p + O(||a||)^3$$
(34)

where $\bar{y}_t = \frac{(1+\eta_{LL})}{(\eta_{cc}+\eta_{LL})} \bar{z}_t$ denotes the level of output when there are no real disturbances and all shocks are set at their means.

Similarly, taking a second order approximation of $V(1-L_t)$ around steady state and using the definition $\hat{l}_t = l_t - \bar{l}_t$ yields

$$\int V\left(1 - L_{ft}\right) = -V_L(1 - L)L \int \left(\hat{l}_{ft} + \frac{(1 + \eta_{ll})}{2}\hat{l}_{ft}^2 + (1 + \eta_{ll})\hat{l}_{ft}\bar{l}_t\right) + t.i.p,$$
(35)

Substituting out l_{ft} using the production function gives

=

$$\int V\left(1 - L_{ft}\right) = -V_L(1 - L_{ft})L \int \left(\hat{y}_{ft} + \frac{(1 + \eta_{ll})}{2}\left(\hat{y}_{ft}^2 - 2\hat{y}_{ft}\hat{z}_t\right) + (1 + \eta_{ll})\hat{y}_{ft}\bar{l}_t\right) + t.i.p$$

$$-V_L(1-L_t)L(E_f(\hat{y}_{ft}) + \frac{(1+\eta_{ll})}{2}E_f(\hat{y}_{ft}^2)$$
(36)

$$-(1+\eta_{ll})\left(E_f(\hat{y}_{ft})\,\hat{z}_t - E_f(\hat{y}_{ft})\,\bar{l}_t\right)) + t.i.p \tag{37}$$

 24 Since I want to compare my results with that of Ball et al., I follow exactly the same steps as Ball et al when deriving the welfare function. (see also Woodford(2003)).

Defining $E_f(\hat{y}_{ft}) = \int \hat{y}_{ft} df$ and $Var_f(\hat{y}_{ft}) = E_f(\hat{y}_{ft}^2) - E_f(\hat{y}_{ft})^2$, taking a second order approximation of (25) and using the resulting expression to substitute for $E_f(\hat{y}_{ft})$, I obtain

$$\int V(1 - L_{ft}) = -V_L(1 - L)L \left(\begin{array}{c} \hat{y}_t + \frac{(1 + \eta_{ll})}{2} \hat{y}_t^2 + \frac{(\theta^{-1} + \eta_{ll})}{2} Var_f(\hat{y}_{ft}) \\ -(1 + \eta_{ll}) \left(\hat{y}_t \hat{z}_t - \hat{y}_t \bar{l}_t \right) \end{array} \right) + t.i.p$$
(38)

Summing (34) with (38), using the steady-state relations $U_C(C)C = V_L(1-L)L$, $(1-\eta_{cc})\bar{y}_t = (1+\eta_{ll})\bar{l}_t$ and the definition of the natural level of output, I obtain

$$U_t(C) + V(1 - L_t) = -\frac{U_C(C)C}{2} \left(\left(\eta_{cc} + \eta_{ll}\right) \left(\hat{y}_t - \hat{y}_t^N\right)^2 + \left(\theta^{-1} + \eta_{ll}\right) Var_f(\hat{y}_{ft}) \right) + t.i.p$$

Dropping all the hats from the output variables since the point of approximation is the same (\bar{y}_t) , using (26) to replace $Var_f(y_{ft})$, I obtain equation (13) in the text.

		ω	a
Smets and Wouters (2005)	Table 1, p. 167	0.13	0.66
Del-Negro et al. (2007)	Table 1, p. 132	0.17	0.76
Rabanal and Rubio-Ramirez (2005)	Table 1, p.1151	0.17	0.76
Justiniano and Primiceri (2008)	Table 1, p. 40	0.10	0.84

Table 1: Estimates of ω and a from the IC model

		Policy rule coefficients		
	(Relative) welfare loss	ϕ_r	ϕ_{π}	ϕ_y
IC	1.09	1.15	0.64	0.04
SI	1.00	2.53	10.77	0.19
Calvo - GTE	1.07	1.11	1.06	0.02

 Table 2: Optimal Taylor rule

	Standard deviations (%)			(Relative) welfare loss
	$(\pi_t - \pi_{t-1})$	π_t	\tilde{y}_t	
IC; $\phi_{\pi} = 0.64$	0.02	0.05	0.78	1.09
IC; $\phi_{\pi} = 1.06$	0.02	0.04	0.93	1.14

Table 3: The IC model

			Policy rule coefficients		
		(Relative) Welfare Loss	ϕ_r	ϕ_{π}	ϕ_y
SI;	$\gamma=0.20,\omega=0.10$	1.07	1.14	1.30	0.02
Calvo- GTE ,	; $\gamma = 0.20, \omega = 0.25$	1.07	1.11	1.06	0.02

Table 4: The optimal Taylor rule in the SI with $\omega = 0.1$ and Calvo-GTE with $\omega = 0.1$ when all the other factors are the same

		Policy rule coefficients		
	(Relative) welfare loss	ϕ_r	ϕ_{π}	ϕ_y
IC	1.26	0.96	0.40	0.00
SI	1.00	1.18	3.90	0.00
Calvo-GTE	1.09	1.01	0.86	0.00

Table 5: Extreme inflation-targeting policy

True Model	Policy generating models				
	(1)	(2)	(3)	(4)	
	Calvo-GTE	IC; $a = 0.76$	SI	IC; a = 1	
Calvo-GTE	1.07	1.31	1.29	2.37	

Table 6: Performance of policy rules that are optimal in the IC and SI when the true model is the Calvo-GTE



Figure 1: (a) Inflation responses in the Indexed-Calvo (IC) and Calvo (b) Inflation response in the SI. (c) Inflation response in the Calvo-GTE.



Figure 2: Inflation response in the SI when $\omega=0.25~(\gamma=0.05)$

NATIONAL BANK OF BELGIUM - WORKING PAPERS SERIES

- 1. "Model-based inflation forecasts and monetary policy rules" by M. Dombrecht and R. Wouters, *Research Series*, February 2000.
- 2. "The use of robust estimators as measures of core inflation" by L. Aucremanne, *Research Series*, February 2000.
- 3. "Performances économiques des Etats-Unis dans les années nonante" by A. Nyssens, P. Butzen, P. Bisciari, *Document Series*, March 2000.
- 4. "A model with explicit expectations for Belgium" by P. Jeanfils, *Research Series*, March 2000.
- 5. "Growth in an open economy: some recent developments" by S. Turnovsky, Research Series, May 2000.
- 6. "Knowledge, technology and economic growth: an OECD perspective" by I. Visco, A. Bassanini, S. Scarpetta, *Research Series*, May 2000.
- 7. "Fiscal policy and growth in the context of European integration" by P. Masson, *Research Series*, May 2000.
- 8. "Economic growth and the labour market: Europe's challenge" by C. Wyplosz, *Research Series*, May 2000.
- 9. "The role of the exchange rate in economic growth: a euro-zone perspective" by R. MacDonald, *Research Series*, May 2000.
- 10. "Monetary union and economic growth" by J. Vickers, Research Series, May 2000.
- 11. "Politique monétaire et prix des actifs: le cas des Etats-Unis" by Q. Wibaut, *Document Series*, August 2000.
- 12. "The Belgian industrial confidence indicator: leading indicator of economic activity in the euro area?" by J.-J. Vanhaelen, L. Dresse, J. De Mulder, *Document Series*, November 2000.
- 13. "Le financement des entreprises par capital-risque" by C. Rigo, Document Series, February 2001.
- 14. "La nouvelle économie" by P. Bisciari, Document Series, March 2001.
- 15. "De kostprijs van bankkredieten" by A. Bruggeman and R. Wouters, Document Series, April 2001.
- 16. "A guided tour of the world of rational expectations models and optimal policies" by Ph. Jeanfils, *Research Series*, May 2001.
- 17. "Attractive Prices and Euro Rounding effects on inflation" by L. Aucremanne and D. Cornille, *Documents Series*, November 2001.
- 18. "The interest rate and credit channels in Belgium: an investigation with micro-level firm data" by P. Butzen, C. Fuss and Ph. Vermeulen, *Research series*, December 2001.
- 19. "Openness, imperfect exchange rate pass-through and monetary policy" by F. Smets and R. Wouters, *Research series*, March 2002.
- 20. "Inflation, relative prices and nominal rigidities" by L. Aucremanne, G. Brys, M. Hubert, P. J. Rousseeuw and A. Struyf, *Research series*, April 2002.
- 21. "Lifting the burden: fundamental tax reform and economic growth" by D. Jorgenson, *Research series*, May 2002.
- 22. "What do we know about investment under uncertainty?" by L. Trigeorgis, Research series, May 2002.
- 23. "Investment, uncertainty and irreversibility: evidence from Belgian accounting data" by D. Cassimon, P.-J. Engelen, H. Meersman, M. Van Wouwe, *Research series*, May 2002.
- 24. "The impact of uncertainty on investment plans" by P. Butzen, C. Fuss, Ph. Vermeulen, *Research series*, May 2002.
- 25. "Investment, protection, ownership, and the cost of capital" by Ch. P. Himmelberg, R. G. Hubbard, I. Love, *Research series*, May 2002.
- 26. "Finance, uncertainty and investment: assessing the gains and losses of a generalised non-linear structural approach using Belgian panel data", by M. Gérard, F. Verschueren, *Research series*, May 2002.
- 27. "Capital structure, firm liquidity and growth" by R. Anderson, Research series, May 2002.
- 28. "Structural modelling of investment and financial constraints: where do we stand?" by J.- B. Chatelain, *Research series*, May 2002.
- 29. "Financing and investment interdependencies in unquoted Belgian companies: the role of venture capital" by S. Manigart, K. Baeyens, I. Verschueren, *Research series*, May 2002.
- 30. "Development path and capital structure of Belgian biotechnology firms" by V. Bastin, A. Corhay, G. Hübner, P.-A. Michel, *Research series*, May 2002.
- 31. "Governance as a source of managerial discipline" by J. Franks, Research series, May 2002.

- 32. "Financing constraints, fixed capital and R&D investment decisions of Belgian firms" by M. Cincera, *Research series*, May 2002.
- 33. "Investment, R&D and liquidity constraints: a corporate governance approach to the Belgian evidence" by P. Van Cayseele, *Research series*, May 2002.
- 34. "On the Origins of the Franco-German EMU Controversies" by I. Maes, Research series, July 2002.
- 35. "An estimated dynamic stochastic general equilibrium model of the Euro Area", by F. Smets and R. Wouters, *Research series*, October 2002.
- 36. "The labour market and fiscal impact of labour tax reductions: The case of reduction of employers' social security contributions under a wage norm regime with automatic price indexing of wages", by K. Burggraeve and Ph. Du Caju, *Research series*, March 2003.
- 37. "Scope of asymmetries in the Euro Area", by S. Ide and Ph. Moës, *Document series*, March 2003.
- 38. "De autonijverheid in België: Het belang van het toeleveringsnetwerk rond de assemblage van personenauto's", by F. Coppens and G. van Gastel, *Document series*, June 2003.
- 39. "La consommation privée en Belgique", by B. Eugène, Ph. Jeanfils and B. Robert, *Document series*, June 2003.
- 40. "The process of European monetary integration: a comparison of the Belgian and Italian approaches", by I. Maes and L. Quaglia, *Research series*, August 2003.
- 41. "Stock market valuation in the United States", by P. Bisciari, A. Durré and A. Nyssens, *Document series*, November 2003.
- 42. "Modeling the Term Structure of Interest Rates: Where Do We Stand?, by K. Maes, *Research series*, February 2004.
- 43. "Interbank Exposures: An Empirical Examination of System Risk in the Belgian Banking System", by H. Degryse and G. Nguyen, *Research series*, March 2004.
- 44. "How Frequently do Prices change? Evidence Based on the Micro Data Underlying the Belgian CPI", by L. Aucremanne and E. Dhyne, *Research series*, April 2004.
- 45. "Firms' investment decisions in response to demand and price uncertainty", by C. Fuss and Ph. Vermeulen, *Research series*, April 2004.
- 46. "SMEs and Bank Lending Relationships: the Impact of Mergers", by H. Degryse, N. Masschelein and J. Mitchell, *Research series*, May 2004.
- 47. "The Determinants of Pass-Through of Market Conditions to Bank Retail Interest Rates in Belgium", by F. De Graeve, O. De Jonghe and R. Vander Vennet, *Research series*, May 2004.
- 48. "Sectoral vs. country diversification benefits and downside risk", by M. Emiris, *Research series*, May 2004.
- 49. "How does liquidity react to stress periods in a limit order market?", by H. Beltran, A. Durré and P. Giot, *Research series*, May 2004.
- 50. "Financial consolidation and liquidity: prudential regulation and/or competition policy?", by P. Van Cayseele, *Research series*, May 2004.
- 51. "Basel II and Operational Risk: Implications for risk measurement and management in the financial sector", by A. Chapelle, Y. Crama, G. Hübner and J.-P. Peters, *Research series*, May 2004.
- 52. "The Efficiency and Stability of Banks and Markets", by F. Allen, Research series, May 2004.
- 53. "Does Financial Liberalization Spur Growth?" by G. Bekaert, C.R. Harvey and C. Lundblad, *Research series*, May 2004.
- 54. "Regulating Financial Conglomerates", by X. Freixas, G. Lóránth, A.D. Morrison and H.S. Shin, *Research series*, May 2004.
- 55. "Liquidity and Financial Market Stability", by M. O'Hara, Research series, May 2004.
- 56. "Economisch belang van de Vlaamse zeehavens: verslag 2002", by F. Lagneaux, *Document series*, June 2004.
- 57. "Determinants of Euro Term Structure of Credit Spreads", by A. Van Landschoot, *Research series*, July 2004.
- 58. "Macroeconomic and Monetary Policy-Making at the European Commission, from the Rome Treaties to the Hague Summit", by I. Maes, *Research series*, July 2004.
- 59. "Liberalisation of Network Industries: Is Electricity an Exception to the Rule?", by F. Coppens and D. Vivet, *Document series*, September 2004.
- 60. "Forecasting with a Bayesian DSGE model: an application to the euro area", by F. Smets and R. Wouters, *Research series*, September 2004.
- 61. "Comparing shocks and frictions in US and Euro Area Business Cycle: a Bayesian DSGE approach", by F. Smets and R. Wouters, *Research series*, October 2004.

- 62. "Voting on Pensions: A Survey", by G. de Walque, *Research series*, October 2004.
- 63. "Asymmetric Growth and Inflation Developments in the Acceding Countries: A New Assessment", by S. Ide and P. Moës, *Research series*, October 2004.
- 64. "Importance économique du Port Autonome de Liège: rapport 2002", by F. Lagneaux, *Document series*, November 2004.
- 65. "Price-setting behaviour in Belgium: what can be learned from an ad hoc survey", by L. Aucremanne and M. Druant, *Research series*, March 2005.
- 66. "Time-dependent versus State-dependent Pricing: A Panel Data Approach to the Determinants of Belgian Consumer Price Changes", by L. Aucremanne and E. Dhyne, *Research series*, April 2005.
- 67. "Indirect effects A formal definition and degrees of dependency as an alternative to technical coefficients", by F. Coppens, *Research series*, May 2005.
- 68. "Noname A new quarterly model for Belgium", by Ph. Jeanfils and K. Burggraeve, *Research series*, May 2005.
- 69. "Economic importance of the Flemish maritime ports: report 2003", F. Lagneaux, *Document series*, May 2005.
- 70. "Measuring inflation persistence: a structural time series approach", M. Dossche and G. Everaert, *Research series*, June 2005.
- 71. "Financial intermediation theory and implications for the sources of value in structured finance markets", J. Mitchell, *Document series*, July 2005.
- 72. "Liquidity risk in securities settlement", J. Devriese and J. Mitchell, Research series, July 2005.
- 73. "An international analysis of earnings, stock prices and bond yields", A. Durré and P. Giot, *Research series*, September 2005.
- 74. "Price setting in the euro area: Some stylized facts from Individual Consumer Price Data", E. Dhyne, L. J. Álvarez, H. Le Bihan, G. Veronese, D. Dias, J. Hoffmann, N. Jonker, P. Lünnemann, F. Rumler and J. Vilmunen, *Research series*, September 2005.
- 75. "Importance économique du Port Autonome de Liège: rapport 2003", by F. Lagneaux, *Document series,* October 2005.
- 76. "The pricing behaviour of firms in the euro area: new survey evidence, by S. Fabiani, M. Druant, I. Hernando, C. Kwapil, B. Landau, C. Loupias, F. Martins, T. Mathä, R. Sabbatini, H. Stahl and A. Stokman, *Research series*, November 2005.
- 77. "Income uncertainty and aggregate consumption, by L. Pozzi, Research series, November 2005.
- 78. "Crédits aux particuliers Analyse des données de la Centrale des Crédits aux Particuliers", by H. De Doncker, *Document series*, January 2006.
- 79. "Is there a difference between solicited and unsolicited bank ratings and, if so, why?" by P. Van Roy, *Research series*, February 2006.
- 80. "A generalised dynamic factor model for the Belgian economy Useful business cycle indicators and GDP growth forecasts", by Ch. Van Nieuwenhuyze, *Research series*, February 2006.
- "Réduction linéaire de cotisations patronales à la sécurité sociale et financement alternatif" by Ph. Jeanfils, L. Van Meensel, Ph. Du Caju, Y. Saks, K. Buysse and K. Van Cauter, *Document series*, March 2006.
- 82. "The patterns and determinants of price setting in the Belgian industry" by D. Cornille and M. Dossche, *Research series*, May 2006.
- 83. "A multi-factor model for the valuation and risk management of demand deposits" by H. Dewachter, M. Lyrio and K. Maes, *Research series*, May 2006.
- 84. "The single European electricity market: A long road to convergence", by F. Coppens and D. Vivet, *Document series*, May 2006.
- 85. "Firm-specific production factors in a DSGE model with Taylor price setting", by G. de Walque, F. Smets and R. Wouters, *Research series*, June 2006.
- 86. "Economic importance of the Belgian ports: Flemish maritime ports and Liège port complex report 2004", by F. Lagneaux, *Document series*, June 2006.
- 87. "The response of firms' investment and financing to adverse cash flow shocks: the role of bank relationships", by C. Fuss and Ph. Vermeulen, *Research series*, July 2006.
- 88. "The term structure of interest rates in a DSGE model", by M. Emiris, Research series, July 2006.
- 89. "The production function approach to the Belgian output gap, Estimation of a Multivariate Structural Time Series Model", by Ph. Moës, *Research series*, September 2006.
- 90. "Industry Wage Differentials, Unobserved Ability, and Rent-Sharing: Evidence from Matched Worker-Firm Data, 1995-2002", by R. Plasman, F. Rycx and I. Tojerow, *Research series,* October 2006.

- 91. "The dynamics of trade and competition", by N. Chen, J. Imbs and A. Scott, *Research series,* October 2006.
- 92. "A New Keynesian Model with Unemployment", by O. Blanchard and J. Gali, *Research series*, October 2006.
- 93. "Price and Wage Setting in an Integrating Europe: Firm Level Evidence", by F. Abraham, J. Konings and S. Vanormelingen, *Research series*, October 2006.
- 94. "Simulation, estimation and welfare implications of monetary policies in a 3-country NOEM model", by J. Plasmans, T. Michalak and J. Fornero, *Research series*, October 2006.
- 95. "Inflation persistence and price-setting behaviour in the euro area: a summary of the Inflation Persistence Network evidence ", by F. Altissimo, M. Ehrmann and F. Smets, *Research series*, October 2006.
- 96. "How Wages Change: Micro Evidence from the International Wage Flexibility Project", by W.T. Dickens, L. Goette, E.L. Groshen, S. Holden, J. Messina, M.E. Schweitzer, J. Turunen and M. Ward, *Research series*, October 2006.
- 97. "Nominal wage rigidities in a new Keynesian model with frictional unemployment", by V. Bodart, G. de Walque, O. Pierrard, H.R. Sneessens and R. Wouters, *Research series*, October 2006.
- 98. "Dynamics on monetary policy in a fair wage model of the business cycle", by D. De la Croix, G. de Walque and R. Wouters, *Research series*, October 2006.
- 99. "The kinked demand curve and price rigidity: evidence from scanner data", by M. Dossche, F. Heylen and D. Van den Poel, *Research series*, October 2006.
- 100. "Lumpy price adjustments: a microeconometric analysis", by E. Dhyne, C. Fuss, H. Peseran and P. Sevestre, *Research series*, October 2006.
- 101. "Reasons for wage rigidity in Germany", by W. Franz and F. Pfeiffer, Research series, October 2006.
- 102. "Fiscal sustainability indicators and policy design in the face of ageing", by G. Langenus, *Research series*, October 2006.
- 103. "Macroeconomic fluctuations and firm entry: theory and evidence", by V. Lewis, *Research series*, October 2006.
- 104. "Exploring the CDS-Bond Basis" by J. De Wit, Research series, November 2006.
- 105. "Sector Concentration in Loan Portfolios and Economic Capital", by K. Düllmann and N. Masschelein, *Research series*, November 2006.
- 106. "R&D in the Belgian Pharmaceutical Sector", by H. De Doncker, *Document series*, December 2006.
- 107. "Importance et évolution des investissements directs en Belgique", by Ch. Piette, *Document series*, January 2007.
- 108. "Investment-Specific Technology Shocks and Labor Market Frictions", by R. De Bock, *Research series*, February 2007.
- 109. "Shocks and frictions in US Business cycles: a Bayesian DSGE Approach", by F. Smets and R. Wouters, *Research series*, February 2007.
- "Economic impact of port activity: a disaggregate analysis. The case of Antwerp", by F. Coppens, F. Lagneaux, H. Meersman, N. Sellekaerts, E. Van de Voorde, G. van Gastel, Th. Vanelslander, A. Verhetsel, *Document series*, February 2007.
- 111. "Price setting in the euro area: some stylised facts from individual producer price data", by Ph. Vermeulen, D. Dias, M. Dossche, E. Gautier, I. Hernando, R. Sabbatini, H. Stahl, *Research series,* March 2007.
- 112. "Assessing the Gap between Observed and Perceived Inflation in the Euro Area: Is the Credibility of the HICP at Stake?", by L. Aucremanne, M. Collin, Th. Stragier, *Research series*, April 2007.
- 113. "The spread of Keynesian economics: a comparison of the Belgian and Italian experiences", by I. Maes, *Research series,* April 2007.
- 114. "Imports and Exports at the Level of the Firm: Evidence from Belgium", by M. Muûls and M. Pisu, *Research series,* May 2007.
- 115. "Economic importance of the Belgian ports: Flemish maritime ports and Liège port complex report 2005", by F. Lagneaux, *Document series*, May 2007.
- 116. "Temporal Distribution of Price Changes: Staggering in the Large and Synchronization in the Small", by E. Dhyne and J. Konieczny, *Research series,* June 2007.
- 117. "Can excess liquidity signal an asset price boom?", by A. Bruggeman, Research series, August 2007.
- 118. "The performance of credit rating systems in the assessment of collateral used in Eurosystem monetary policy operations", by F. Coppens, F. González and G. Winkler, *Research series*, September 2007.
- 119. "The determinants of stock and bond return comovements", by L. Baele, G. Bekaert and K. Inghelbrecht, *Research series*, October 2007.

- 120. "Monitoring pro-cyclicality under the capital requirements directive: preliminary concepts for developing a framework", by N. Masschelein, *Document series*, October 2007.
- 121. "Dynamic order submission strategies with competition between a dealer market and a crossing network", by H. Degryse, M. Van Achter and G. Wuyts, *Research series*, November 2007.
- 122. "The gas chain: influence of its specificities on the liberalisation process", by C. Swartenbroekx, *Document series*, November 2007.
- 123. "Failure prediction models: performance, disagreements, and internal rating systems", by J. Mitchell and P. Van Roy, *Research series*, December 2007.
- 124. "Downward wage rigidity for different workers and firms: an evaluation for Belgium using the IWFP procedure", by Ph. Du Caju, C. Fuss and L. Wintr, *Research series*, December 2007.
- 125. "Economic importance of Belgian transport logistics", by F. Lagneaux, Document series, January 2008.
- 126. "Some evidence on late bidding in eBay auctions", by L. Wintr, Research series, January 2008.
- 127. "How do firms adjust their wage bill in Belgium? A decomposition along the intensive and extensive margins", by C. Fuss, *Research series*, January 2008.
- 128. "Exports and productivity comparable evidence for 14 countries", by The International Study Group on Exports and Productivity, *Research series*, February 2008.
- 129. "Estimation of monetary policy preferences in a forward-looking model: a Bayesian approach", by P. Ilbas, *Research series*, March 2008.
- 130. "Job creation, job destruction and firms' international trade involvement", by M. Pisu, *Research series*, March 2008.
- 131. "Do survey indicators let us see the business cycle? A frequency decomposition", by L. Dresse and Ch. Van Nieuwenhuyze, *Research series*, March 2008.
- 132. "Searching for additional sources of inflation persistence: the micro-price panel data approach", by R. Raciborski, *Research series*, April 2008.
- 133. "Short-term forecasting of GDP using large monthly datasets A pseudo real-time forecast evaluation exercise", by K. Barhoumi, S. Benk, R. Cristadoro, A. Den Reijer, A. Jakaitiene, P. Jelonek, A. Rua, G. Rünstler, K. Ruth and Ch. Van Nieuwenhuyze, *Research series*, June 2008.
- 134. "Economic importance of the Belgian ports: Flemish maritime ports, Liège port complex and the port of Brussels report 2006" by S. Vennix, *Document series*, June 2008.
- 135. "Imperfect exchange rate pass-through: the role of distribution services and variable demand elasticity", by Ph. Jeanfils, *Research series*, August 2008.
- 136. "Multivariate structural time series models with dual cycles: Implications for measurement of output gap and potential growth", by Ph. Moës, *Research series*, August 2008.
- 137. "Agency problems in structured finance a case study of European CLOs", by J. Keller, *Document series*, August 2008.
- 138. "The efficiency frontier as a method for gauging the performance of public expenditure: a Belgian case study", by B. Eugène, *Research series*, September 2008.
- 139. "Exporters and credit constraints. A firm-level approach", by M. Muûls, *Research series*, September 2008.
- 140. "Export destinations and learning-by-exporting: Evidence from Belgium", by M. Pisu, *Research series*, September 2008.
- 141. "Monetary aggregates and liquidity in a neo-Wicksellian framework", by M. Canzoneri, R. Cumby, B. Diba and D. López-Salido, *Research series*, October 2008.
- 142 "Liquidity, inflation and asset prices in a time-varying framework for the euro area, by Ch. Baumeister, E. Durinck and G. Peersman, *Research series*, October 2008.
- 143. "The bond premium in a DSGE model with long-run real and nominal risks", by Glenn D. Rudebusch and Eric T. Swanson, *Research series*, October 2008.
- 144. "Imperfect information, macroeconomic dynamics and the yield curve: an encompassing macro-finance model", by H. Dewachter, *Research series*, October 2008.
- 145. "Housing market spillovers: evidence from an estimated DSGE model", by M. Iacoviello and S. Neri, *Research series*, October 2008.
- 146. "Credit frictions and optimal monetary policy", by V. Cúrdia and M. Woodford, *Research series*, October 2008.
- 147. "Central Bank misperceptions and the role of money in interest rate rules", by G. Beck and V. Wieland, *Research series*, October 2008.
- 148. "Financial (in)stability, supervision and liquidity injections: a dynamic general equilibrium approach", by G. de Walque, O. Pierrard and A. Rouabah, *Research series*, October 2008.

- 149. "Monetary policy, asset prices and macroeconomic conditions: a panel-VAR study", by K. Assenmacher-Wesche and S. Gerlach, *Research series*, October 2008.
- 150. "Risk premiums and macroeconomic dynamics in a heterogeneous agent model", by F. De Graeve, M. Dossche, M. Emiris, H. Sneessens and R. Wouters, *Research series*, October 2008.
- 151. "Financial factors in economic fluctuations", by L. J. Christiano, R. Motto and M. Rotagno, *Research series*, to be published.
- 152. "Rent-sharing under different bargaining regimes: Evidence from linked employer-employee data", by M. Rusinek and F. Rycx, *Research series*, December 2008.
- 153. "Forecast with judgment and models", by F. Monti, Research series, December 2008.
- 154. "Institutional features of wage bargaining in 23 European countries, the US and Japan", by Ph. Du Caju, E. Gautier, D. Momferatou and M. Ward-Warmedinger, *Research series*, December 2008.
- 155. "Fiscal sustainability and policy implications for the euro area", by F. Balassone, J. Cunha, G. Langenus, B. Manzke, J Pavot, D. Prammer and P. Tommasino, *Research series*, January 2009.
- 156. "Understanding sectoral differences in downward real wage rigidity: workforce composition, institutions, technology and competition", by Ph. Du Caju, C. Fuss and L. Wintr, *Research series*, February 2009.
- 157. "Sequential bargaining in a New Keynesian model with frictional unemployment and staggered wage negotiation", by G. de Walque, O. Pierrard, H. Sneessens and R. Wouters, *Research series*, February 2009.
- 158. "Economic Importance of Air Transport and Airport Activities in Belgium", by F. Kupfer and F. Lagneaux, *Document series*, March 2009.
- 159. "Rigid labour compensation and flexible employment? Firm-Level evidence with regard to productivity for Belgium", by C. Fuss and L. Wintr, *Research series*, March 2009.
- 160. "The Belgian Iron and Steel Industry in the International Context", by F. Lagneaux and D. Vivet, *Document series*, March 2009.
- 161. "Trade, wages and productivity", by K. Behrens, G. Mion, Y. Murata and J. Südekum, *Research series*, March 2009.
- 162. "Labour flows in Belgium", by P. Heuse and Y. Saks, Research series, April 2009.
- 163. "The young Lamfalussy: an empirical and policy-oriented growth theorist", by I. Maes, *Research series*, April 2009.
- 164. "Inflation dynamics with labour market matching: assessing alternative specifications", by K. Christoffel, J. Costain, G. de Walque, K. Kuester, T. Linzert, S. Millard and O. Pierrard, *Research series*, May 2009.
- 165. "Understanding inflation dynamics: Where do we stand?", by M. Dossche, Research series, June 2009.
- 166. "Input-output connections between sectors and optimal monetary policy", by E. Kara, *Research series*, June 2009.
- 167. "Back to the basics in banking? A micro-analysis of banking system stability", by O. De Jonghe, *Research series*, June 2009.
- 168. "Model misspecification, learning and the exchange rate disconnect puzzle", by V. Lewis and A. Markiewicz, *Research series*, July 2009.
- 169. "The use of fixed-term contracts and the labour adjustment in Belgium", by E. Dhyne and B. Mahy, *Research series*, July 2009.
- 170. "Analysis of business demography using markov chains An application to Belgian data", by F. Coppens and F. Verduyn, *Research series*, July 2009.
- 171. "A global assessment of the degree of price stickiness results from the NBB business survey", by E. Dhyne, *Research series*, July 2009.
- 172. "Economic importance of the Belgian ports: Flemish maritime ports, Liège port complex and the port of Brussels report 2007" by C. Mathys, *Document series*, July 2009.
- 173. "Evaluating a monetary business cycle model with unemployment for the euro area", by N. Groshenny, *Research series*, July 2009.
- 174. "How are firms' wages and prices linked: survey evidence in Europe", by M. Druant, S. Fabiani, G. Kezdi, A. Lamo, F. Martins and R. Sabbatini, *Research series*, August 2009.
- 175. "Micro-data on nominal rigidity, inflation persistence and optimal monetary policy", by E. Kara, *Research series*, September 2009.